%% filename: cite-xs.tex
%% version: 1.00
%% date: 2004/06/30
%%
%% American Mathematical Society
%% Technical Support
%% Publications Technical Group
%% 201 Charles Street
%% Providence, RI 02904
%% USA
%% tel: (401) 455-4080
%%      (800) 321-4267 (USA and Canada only)
%% fax: (401) 331-3842
%% email: tech-support@ams.org
%% 
%% Copyright 2004, 2010 American Mathematical Society.
%% 
%% This work may be distributed and/or modified under the
%% conditions of the LaTeX Project Public License, either version 1.3c
%% of this license or (at your option) any later version.
%% The latest version of this license is in
%%   http://www.latex-project.org/lppl.txt
%% and version 1.3c or later is part of all distributions of LaTeX
%% version 2005/12/01 or later.
%% 
%% This work has the LPPL maintenance status `maintained'.
%% 
%% The Current Maintainer of this work is the American Mathematical
%% Society.
%%
%% ====================================================================

\documentclass{amsart}
\usepackage{amsrefs}

\begin{document}

\section{Introduction}

A single cite \cite{MR1451594}
and a multiple \cites{Weyl,Gould,Rob,MR95f:17013,Ser77}.
And then with an option in the middle:
\citelist{\cite{Rob} \cite{Gould} \cite{MR95f:17013}*{Chapter 18}}.

\begin{bibsection}
\begin{biblist}
\bibselect{gktest}
\end{biblist}
\end{bibsection}

\end{document}
